Author's response to reviews

Title: MAVIDOS Maternal Vitamin D Osteoporosis Study: study protocol for a randomized controlled trial.

Authors:

Nicholas C Harvey MD (nch@mrc.soton.ac.uk)
Kassim Javaid MD (kassim.javaid@ndorms.ox.ac.uk)
Nicholas Bishop MD (n.j.bishop@sheffield.ac.uk)
Stephen Kennedy MD (stephen.kennedy@obs-gyn.ox.ac.uk)
Aris Papageorghiou MD (apapageo@sgul.ac.uk)
Robert Fraser MD (Robert.Fraser@sth.nhs.uk)
Saurabh V Gandhi MD (Saurabh.Gandhi@sth.nhs.uk)
Inez Schoenmakers PhD (inez.schoenmakers@mrc-hnr.cam.ac.uk)
Ann Prentice PhD (ann.prentice@mrc-hnr.cam.ac.uk)
Cyrus Cooper FMedSci (cc@mrc.soton.ac.uk)

Version: 7 Date: 13 January 2012

Author's response to reviews: see over
Dear Professor Altman,

Thank you very much for accepting our paper for publication in *Trials*.

We have corrected the article title as per the editorial request below:

? Title: please ensure the title conforms to journal style for study protocol articles. The title should follow the format ?_________: study protocol for a randomized controlled trial?.

Please do not hesitate to contact me if you require any further information.

Yours sincerely,

Dr Nicholas C Harvey MA MB BChir MRCP PhD
Senior Lecturer and Honorary Consultant Rheumatologist